Author's response to reviews

Title: The Stigma of Obesity in the General Public and its Implications for Public Health - a Systematic Review

Authors:

- Claudia Sikorski (Claudia.Sikorski@medizin.uni-leipzig.de)
- Melanie Luppa (Melanie.Luppa@medizin.uni-leipzig.de)
- Marie Kaiser (Marie.Kaiser@medizin.uni-leipzig.de)
- Heide Glaesmer (Heide.Glaesmer@medizin.uni-leipzig.de)
- Georg Schomerus (georg.schomerus@uni-greifswald.de)
- Hans-Helmut König (h.koenig@uke.uni-hamburg.de)
- Steffi G Riedel-Heller (Steffi.Riedel-Heller@medizin.uni-leipzig.de)

Version: 4 Date: 16 August 2011

Author's response to reviews: see over
Dear Editors,

Please find enclosed the corrected version of the manuscript entitled “The Stigma of Obesity in the General Public and its Implications for Public Health - a Systematic Review”.

To finalize the manuscript for publication we re-ran a spelling check and corrected the found typos. Following your instructions, we made the requested change regarding the additional file (see page 23). As for minor changes, we removed all highlighting and all vertical lines in the tables. We also re-formatted the tables, so we do not have any merged cells now. Additionally, we formatted the Figure we have included in the manuscript. It is now in WORD format and displays much more detail. We changed the font to Arial as proposed.

Organizational process for payment of the page charges is underway so we hope that the manuscript is ready for publication in the near future.

Yours sincerely,

Claudia Sikorski